Author's response to reviews

Title: Payments and quality of care in private for-profit and public hospitals in Greece

Authors:

Elias Kondilis (eliaskon@med.auth.gr)
Magda Gavana (gavana@med.auth.gr)
Stathis Giannakopoulos (gianstat@med.auth.gr)
Emmanouil Smyrnakis (smyrnak@auth.gr)
Nikolaos Dombros (dombros@auth.gr)
Alexis Benos (benos@med.auth.gr)

Version: 5 Date: 29 August 2011

Author's response to reviews:

Dear editor,

we send you a revised copy of our paper that follows the structure of Research Articles published at the BMC Health Services Research.

Since the first draft of our paper was written using the relevant template that appears at the Journal's website, only few changes had to be made concerning at least the structure of our manuscript:

1. we added zip codes at the authors' addresses
2. we added the section "Acknowledgements and Funding" stating that none of the authors of the study received any funding for the research and the preparation of the manuscript
3. we added the section "Authors’ information" rewriting authors’ affiliations as stated at the title page of the manuscript.

Yours sincerely

Dr Elias Kondilis

Corresponding author